Author’s response to reviews

Title: What Drives Junior Doctors To Use Clinical Practice Guidelines? A National Cross-Sectional Survey Of Foundation Doctors In England & Wales

Authors:

Logan Manikam (logan.manikam.10@ucl.ac.uk)
Andrew Hoy (andrewrh.work@gmail.com)
Hannah Fosker (hannahfosker@nhs.net)
Martin Wong (martin.wong@doctors.org.uk)
Jay Banerjee (jb234@le.ac.uk)
Monica Lakhanpaul (m.lakhanpaul@ucl.ac.uk)
Alec Knight (alec.knight@kcl.ac.uk)
Peter Littlejohns (peter.littlejohns@kcl.ac.uk)

Version: 2 Date: 03 Dec 2015

Author’s response to reviews:

1. Tables have been uploaded as separate files.

2. Previous versions of the manuscripts have been removed

3. The funding section has been incorporated into the acknowledgements